

Contents lists available at ScienceDirect

World Patent Information



journal homepage: www.elsevier.com/locate/worpatin

Responsible intellectual property strategy for sustainability transition - An exploratory study $\stackrel{\star}{\times}$



Pratheeba Vimalnath^{a,*}, Frank Tietze^a, Elisabeth Eppinger^b, Akriti Jain^c, Anjula Gurtoo^d, Maximilian Elsen^a

^a Department of Engineering, Innovation and Intellectual Property Management Laboratory, Centre for Technology Management, Institute for Manufacturing, University of Cambridge, CB3 0FS, UK

^b HTW Berlin, Campus Wilhelminenhof, WH Gebäude A, Raum 417, Wilhelminenhofstraße 75A, 12459, Berlin, Germany

^c International Management Institute - New Delhi, B-10, Qutab Institutional Area, Tara Crescent, New Delhi, Delhi, 110016, India

^d Centre for Society and Policy, Indian Institute of Science, Bengaluru, 560012, Karnataka, India

ARTICLE INFO

Keywords: IP strategy Responsible IP strategy Responsible innovation Sustainability Sustainable development

ABSTRACT

In this paper, we propose a responsible intellectual property (IP) strategy (R-IPS) framework based on five exploratory case studies of sustainable companies in energy, nutrition, consumer electronics, manufacturing and water treatment sectors. These companies responsibly use IP assets to create positive social and environmental impact (or reduce negative impact), and unlock new opportunities for financial (economic) gains. The extent to which firms today use IP rights such as patents and trademarks, and IP more broadly (including informal intellectual assets such as data and trade secrets, and contracts) as a strategic tool to facilitate sustainability remains to be understood better. More specifically, we need understanding of how companies could, or even should manage and use their IP more responsibly in these changing times during which transitions towards sustainable development are so thoroughly and urgently needed. The proposed framework defined using five dimensions namey deliberation/intention, sustainability IP alignment, flexibility and timing, inclusiveness, and co-creation can support managerial decision-making in formulating or re-designing IP strategies to increase organization's social and environmental impact, maximising their contributions (e.g. accelerating transitions) towards building sustainable economies.

1. Introduction

To address global challenges, such as decarbonisation to achieve Net Zero, we need large-scale research programmes to develop and quickly diffuse sustainable innovations. Since the United Nation's Sustainable Development Goals (SDGs) launch in 2015, worldwide innovations ranging from technological and business models to policy-oriented ones thrive to address global issues such as climate change, poverty, inequality, and other SDGs [1–3]. Recently, the COVID-19 pandemic has highlighted and demonstrated the possibility of increasing the pace of research, development, and large-scale diffusion of innovations to address urgent global challenges [4]. While during the pandemic much

emphasis was on social-impact innovations (e.g. vaccines and personal protective equipment), we also need 'green innovations' [5] to address climate change and achieve decarbonisation goals. Those include, for instance, more environment-friendly technologies, products, services, and combinations thereof. Furthermore, we also need system-changing innovations, policy innovations and innovative business models that aim to mitigate environmental risks, such as through pollution control, emission reduction, energy-saving and waste recycling [6–8]. Digital (e. g. AI-based) sustainable innovations are likely to play a particular role for sustainability [9]. For instance, while Google has always been on the forefront of optimising its data centre's energy consumption, 'unleashing' Deepmind onto this problem has resulted in an additional energy

https://doi.org/10.1016/j.wpi.2023.102195

Received 31 January 2022; Received in revised form 21 December 2022; Accepted 27 March 2023 Available online 15 April 2023

0172-2190/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Frank Tietze and Anjula Gurtoo reports financial support was provided by Economic and Social Research Council, United Kingdom. Elisabeth Eppinger reports financial support was provided by Federal Ministry of Education and Research Bonn Office, Germany. One of the co-authors of this paper is a guest editor to the special issue for which this paper is submitted.

^{*} Corresponding author.

E-mail addresses: pv302@cam.ac.uk (P. Vimalnath), frank.tietze@eng.cam.ac.uk (F. Tietze), Elisabeth.Eppinger@HTW-Berlin.de (E. Eppinger), akriti.jain@imi. edu (A. Jain), anjula@iisc.ac.in (A. Gurtoo), mhce2@cam.ac.uk (M. Elsen).

reduction from what was thought to be already a heavily optimised system by about 40% [10].

Sustainability-oriented innovations (referred to hereafter as SOI) [11,12] expand the scope of innovation to include environmental as well as social and economic dimensions. However, several of these innovations involve complex intellectual property (IP) related challenges that potentially hinder their development and diffusion [13,14]. For instance, when the IP assets required for developing an invention with critical environmental or social benefit (e.g. COVID vaccine) have multiple organizational ownership, and the organizations are unwilling to share the 'crisis-critical' IP or license to an external entity for mass-producing the invention, then social benefit gets delayed or hindered (e.g. delayed vaccine development and diffusion in the case of COVID costing human lives) [4,15,16]. Another example, related to the circular economy are certain platform technologies, for which all actors need access to the relevant IP to jointly 'close' the circular economy loops. Such technologies might be called 'common purpose' technologies with associated 'common good IP'.

By IP, in this paper we refer to formal IP rights (IPR) like patents, trademark, and copyrights, and informal intellectual assets like knowhow and data [17]. IPR has been an established policy instrument to incentivise innovation but using IP (including IPR) as a strategic tool to facilitate sustainability remains little understood. Despite the strong dependencies between IP and innovation, literature lacks an understanding of how companies should responsibly manage and use IP to govern innovations, especially technologies potentially supporting SDGs.

Based on case study evidence from five companies that have successfully developed and diffused sustainable innovations (referred to hereafter as sustainable companies), this paper provides insights into how these companies have used their IP responsibly to contribute towards sustainability. We identify, define, and describe five dimensions companies should consider when formulating responsible IP strategies, such as for helping to achieve SDGs. By IP strategy, we refer to the decision-making guidance of an organisation regarding the selection and combination of different 'IP models' to maximise value creation and capture in support of, and thus in alignment with the organisation's objectives. The notion of an 'IP model' refers to how an organisation controls the ownership, access and usage rights for a combination of relevant IP assets to achieve one or more specific objectives [18,19]. The choice of the IP models might very well depend on the actors, activities, artifacts, and institutions, and relations of the organization within and beyond its ecosystem [20].

We incorporate learnings from the responsible research and innovation (RRI) literature to understand an organization's strategic, but responsible use of IP. The theory of RRI provides guidelines for responsibly governing science and innovations through an interactive approach among relevant stakeholders taking into account societal expectations and future developments [21–23] but has not incorporated IP considerations, despite mentioning that IPR needs to be addressed [23]. Our investigation of the IP strategies of the five sustainable case companies studied as a part of the IPACST¹ project indicates the responsible use of IP for sustainability going beyond the traditional protective mindset and places emphasis on IP sharing for common good. This approach is evidenced commonly in the open innovation context, mainly using IP to trigger and create (if not govern) change towards sustainability within the company's boundaries and across ecosystem stakeholders. Having identified the five dimensions we define an organizationallevel, responsible IP strategy as an "IP strategy that deliberately aligns with the organization's sustainability mission; is inclusive (in terms of IP sharing) of sustainability promoting stakeholders; and spans the boundary to enable knowledge co-creation, such as for common purpose technologies. Such IP strategy considers the different impact dimensions (e.g. social, environmental, and economic implications) and societal expectations (e.g. in its value definition) to flexibly evolve over time to best support long-term sustainability transitions."

The paper is structured as follows. Section 2 provides the theoretical background about IP strategy from three viewpoints: IP strategy from the view of IP as a business asset, IP strategy for sustainable development, and IP in times of institutional changes. Section 3 explains the methodology used, including sampling, data collection, triangulation, and analysis. Section 4 presents the results describing and discussing the dimensions derived for responsible IP strategy, and section 5 concludes the paper with implications, limitations, and future research directions.

2. Theoretical background

2.1. IP strategy and IP as a business asset

IP strategies aim to protect and increase the competitive advantage of businesses (e.g., [24–26]). The traditional incentive theory of IPR argues that IPR systems incentivise innovators and can act as a strategic tool to attract investments for businesses to grow [27], build competitive advantage [28], recoup R&D investments by inventors including smaller firms such as start-ups by internal use [25], and in collaborations with strategic partners [29,30].

Although IPR and IP strategies are usually discussed and analysed in the innovation and technology management contexts, IP strategies go beyond innovation protection. Some IP protection instruments such as patents and utility models focus on protecting technological innovation. While other IP protection instruments such as trademarks, trade secrets and informal intellectual assets such as know-how, data protection strategies and lead time advantages cover business activities and business models beyond innovation efforts more broadly. As part of the intellectual capital of businesses, IP and IPR have become significant assets - in some instances, more valuable than the physical property of businesses [31,32].

More recently, critical economic analysis discusses IPR as strategic tools to manifest and increase the power of businesses and by this deincentivise companies to put efforts into and compete for productivity growth [33]. When analysing innovation from a broader economic perspective and not from a single firm, especially patents were criticised for holding up innovation given the cumulative nature of innovation where a product or technology combines numerous inventions spread across different owners [34,35].

2.2. Role of IP for sustainability

The speed with which countries move towards sustainability depends (at least to some extent) on their technological development, knowledge accumulation, access, and absorption of green and sustainable technologies [36–38]. However, owing to the complex nature of sustainable technologies [13,36,39,40], the body of literature on IP for sustainability presents conflicting views.

One group of researchers suggests IPRs such as patents, trademarks, copyrights, and databases form an organisations' structural capital [41], enhancing an organization's sustainable innovation process [42] and environment-related activities [43]. Researchers in this stream suggest IPR as an instrument to facilitate technology transfer for sustainable development [44]. During the early phases of industry development for clean technologies, major emerging countries like China and India benefited tremendously through inward patent licensing of solar photovoltaic and wind technology [32]. From a social sustainability

¹ Intellectual Property Models for Accelerating Sustainability Transitions (IPACST) (www.ip4sustainability.org) is a four-year international and interdisciplinary research project for understanding the role played by intellectual property (IP) in accelerating sustainability transitions. IPACST is one of 12 transnational research projects by the Belmont Forum and NORFACE joint programme on Transformations to Sustainability (T2S).

perspective, a stream of research shows firm's collaborative IP generations and ownership positively influencing social sustainability [45–47].

However, another set of researchers argues for IPR hindering sustainable development. For instance, from a technology perspective, IP provides a weak incentive to develop green technologies due to a longtime gap between green technological invention and their first commercialisation [36]. Moreover, since many companies hold patent rights on different complex technical components, licensing fees may stake up, increasing transaction cost, thus, blocking the diffusion of green technologies [14,36]. Thus, a protective approach to IP can hinder sustainable development if IPR owners block others from accessing IP protected technologies [48,49]. In environmental and social sustainability contexts, theories around openness and sharing of IP through wider sharing mechanisms like patent commons, pledges and open-source [8,38,50] are discussed as alternatives.

2.3. IP in times of institutional changes

Organisations have long used IP to strategically adapt to structural changes in the economy or respective industries [30,51,52] during normal economic development situations but also during crises. The literature on strategic IP management provides examples of companies using IP to push structural changes in their industry [30,51,52], for example, in the agriculture industry [52]. In the ongoing COVID-19 pandemic, IP has played a role in facilitating rapid manufacturing of crisis-critical products (such as ventilators, masks, sanitisers). IP access has been granted (or rather IP rights have not been enforced) to companies from the non-medical sector that have capacity to manufacture crisis-critical products, enabling them to manufacture relevant products and equipment available for urgent and critical needs [4,15].

Business model innovation forms another significant factor for changing industry structures and innovation capabilities. Business models often rely on IP and are centred around the development and licensing of general-purpose technologies, pushing institutional changes in return [51]. In the context of sustainability, however, we find the existing literature lacking a structured understanding of how IP can be used in times of change.

2.4. Theory of responsible research & innovation (RRI)

The theory of responsible research and innovations (RRI) has been widely discussed to achieve SDGs. RRI encompasses more inclusive concepts for science and innovation processes, including addressing the needs of less affluent communities, anticipating possible negative effects of innovation, acknowledging tipping points, and limited natural resources. Von Schomberg [22] defines RRI as 'a transparent, interactive process by which societal actors and innovators become mutually responsive to each other with a view to the (ethical) acceptability, sustainability and societal desirability of the innovation process and the marketable products (to allow a proper embedding of scientific and technological advances in our society)'.

What does RRI mean in practice? Stilgoe et al. [21] define RRI as 'taking care of the future through collective stewardship of science and innovation in the present' (p 1570) and identify four dimensions for integration in research and innovation, namely, reflexivity, anticipation, inclusiveness/deliberation, and responsiveness. Reflexivity combines three factors for reflection: underlying purpose, motivation and potential impact; anticipation consists of two factors for impact analysis: creating appropriate solutions and creating appropriate policies; inclusiveness refers to listening to the public and the stakeholders; and responsiveness refers to using reflexivity, anticipation, and inclusiveness to set the direction of research and innovation.

While no clear definition of responsible IP strategy exists, two discussions in the literature are relevant: open approaches as an appropriate response for responsible IP and human rights as the responsible IP perspective. For the first one, for instance, Konig et al. [53] study open

IP in synthetic biology and argue against open IP. The authors state a more appropriate question to ask for responsible IP would be how patents and property rights are applied when fostering the development and accessibility of innovations for social benefit. Collaborations with non-profit organisations [54,55] and technology transfer to lower-income countries [56] are some options suggested by Konig et al. [53] as well. Beyer [57], looking at public health, argues for non-exclusive licensing. The author lists characteristics of the sharing agreement, terms and conditions of the licence, including territory, royalties and technology transfer modalities, as some of the main indicators of responsible IP.

Second, research focusing on the relations between human rights and IP appears relevant for understanding what responsible IP could mean. Brown [58] argues for influencing corporate conduct based on human rights and making human rights the basis for legal enforcement. What would be the ambit of those obligations and to what limit should the balancing act be conducted are answered by Brown [58] by discussing options of compulsory conduct, voluntary conduct and use of CSR to meet human rights obligations of IP.

While participatory collaboration processes for more inclusive research, innovation, and technology transfer, require decision-making about sharing mechanisms, joint development and exchange of IP, the empirical and conceptual research on RRI lacks evidence and advice on good practice examples. Stilgoe et al. [21] point towards a need to rethink IPR and patents to move toward RRI; so far, existing research hardly guides adequate management and use of IP. Existing literature lacks evidence-based results to define, characterise and formulate responsible IP strategies for building a sustainable economy. Accordingly, literature lacks consensus on the strategic management and responsible use of IP that best supports sustainability.

3. Methodology

The study adopts an exploratory research design, given a limited theoretical and empirical understanding of IP strategies of sustainable companies and dimensions to define responsible IP strategies. Adopting a comparative, in-depth case study approach [59,60], the study develops and analyses five companies with proven social and/or environmental sustainability impact to define and characterise responsible IP strategies for sustainability. Given the limited prior research in this field, comparative case studies allow for an in-depth analysis to explore phenomena and theory building for further investigations. While the external validity of qualitative case studies is generally lower than for large-scale quantitative studies, it is increased by comparing cases in different settings, i.e. regions, countries and sectors [59,60].

The case companies were target selected based on a set of pre-defined criteria. These include the company should have (i) at least one sustainability-oriented innovation (SOI) (technology, product or service) as the core business value proposition, (ii) an IP strategy managing the ownership and the usage of IP assets relevant to the company's SOI and (iii) a proven sustainability impact recognized by any of the united nation bodies or sustainability awards. We adopted a carefully designed multi-step triangulated data-collection methodology [60] to overcome potential biases in the sample (arising due to 'greenwashing' [61,62] by companies), and to ensure internal validity of the findings. For this, we followed a three-step process.

First, we identified, shortlisted, and selected companies recognized by one or more notable international awarding bodies (after undergoing a stringent assessment process) for their significant environmental or social contributions. Next, we collected secondary data about the companies from company's annual reports, sustainability reports, business responsibility reports, Global Reporting Initiatives (GRI) reports, press releases and websites, and other platforms such as media news articles and web articles. Finally, to validate and triangulate the data collected and gain first-hand insights into the company's sustainability performance, we interviewed respective executives and top managers, and asked for specific facts supporting the performance indicators. For example, if the company states that they have contributed to community development, then we asked them for the number of community development programs conducted, the number of beneficiaries over years, among other validation questions. With this three-step process, including recognitions by external notable awarding bodies, secondary data from websites (prior to talking to companies), and primary interviews with the relevant company executives, we could validate the company's sustainability claims.

The five case studies selected for the development of the responsible IP strategy framework consist of (1) a mission-driven innovative nutrition solution provider,² (2) a mission-driven renewable energy incumbent, (3) a social enterprise in the consumer electronic/manufacturing sector, (4) an incumbent providing energy-efficient solutions for industrial and commercial applications and (5) a waste-treatment technology development company. In addition, a sixth company, a large multinational (about 130 plus years old) operating earlier as a traditional profit-oriented business and transitioning to sustainability in the past decade, is included for the external validity of the responsible IP strategy dimensions derived. These case studies constitute a sub-set of the sustainable companies studied as a part of the IPACST project.

For each case study, the study uses primary as well as secondary data sources as described earlier to track the evolution of IP strategy, sustainable business model (SBM) [63], and their sustainability impact (internal as well as external). Each case development involves a four-step process. As a first step, credible secondary data sources are used to map the evolution of the case company's IP strategy, SBM, and sustainability impact in a visual mapping template. Secondly, semi-structured in-depth interviews are conducted with relevant company employees (e.g., Chief Technology Officer (CTO)/IP counsel/IP head for IP strategy interview; Chief Executive Officer (CEO)/sustainability head for sustainability impact and business model interview) to gather primary data. The interviews are conducted using an interactive, visual mapping approach. Each case involves a total of four interviews. Each interview lasts between 1 and 1.5 h (about 4-6.5 h of primary interview per case). Thirdly, visual maps are refined after synthesising and corroborating the data captured from secondary and primary sources and any discrepancies clarified with the companies in the second round of interview. This multi-step approach enables comprehensive data capture and validation from reliable and credible sources ensuring high-quality data and internal validity. Fig. 1 presents an overview of the data collection process along with the visual map templates used. Finally, the synthesised data is presented back to the participants for validation and feedback.

We analysed the data using qualitative content analysis methodology, starting with a directed content analysis approach [64] based on theoretical dimensions derived from the theory of RRI. We then followed the iterative hybrid method that uses literature as well as case description [65] to derive dimensions of responsible IP strategy iteratively. The dimensions from the RRI, namely reflexivity, anticipation, inclusiveness, deliberation, and responsiveness, are used as starting points for coding the cases. Those RRI dimensions that are relevant but limited in scope to capture IP relevance have been redefined while developing codes from the case data. In addition, a new code called co-creation emerged from the data analysis. Fig. 2 depicts the final coding scheme. The data analysis resulted in twelve first-order codes aggregated into 5 second-order codes representing the five dimensions of the proposed responsible IP strategy namely deliberation/intention, sustainability IP alignment, flexibility and timing, inclusiveness, and co-creation. These five dimensions are further grouped into two high-level categories namely the intra-firm and the ecosystem-level categories. The intra-firm

level category represents decisions within the firm boundaries, without involving other players, which include three dimensions namely sustainability alignment, deliberation/intention, and flexibility and timing. In contrast, the ecosystem-level category represents actions involving one or more external entities for implementation and includes two dimensions namely inclusiveness, and co-creation.

To improve the external validity and to demonstrate the trustworthiness and rigour of our findings [66,67], the dimensions were applied and compared before and during the transition period of a sixth case company in its sustainability transition period.

4. Results and discussion

Two patterns emerge from the data analysis. First, all the five case studies confirm that a strategic approach to IP forms an integral part of companies' SBM, and companies deliberately design their IP strategies to achieve sustainability goals. Second, the five case companies adopt different IP strategies but exhibit similarities in their considerations in responsibly designing their IP strategies for sustainability. Building on the dimensions of the theory of RRI and the results from the case studies lead to the identification of five dimensions are: (i) deliberation/ intention, (ii) sustainability IP alignment, (iii) flexibility and timing, (iv) inclusiveness, and (v) co-creation (see Fig. 3). Table 1 summarises these five dimensions of the IP strategies adopted by each of the five case studies. Below we explain each dimension with case examples and supporting interview quotes.

4.1. Intra-firm level dimensions

Intra-firm level category includes three dimensions namely deliberation/intention, sustainability alignment, and flexibility/timing dimension.

4.1.1. Deliberation/intention

Responsible IP strategies consider environmental/social implications alongside traditional economic focus by choice and not by chance. This finding confirms prior research stressing transformative changes toward sustainability call for 'deliberate', 'intentional' and 'anticipatory' approaches [68–70]. All five case companies deliberately designed their IP strategies to achieve their sustainability mission by choice and not just comply with regulations or as a part of their corporate social responsibility obligations.

For example, in case #1, the founder started the company as a family business with the mission to prevent and treat malnutrition among vulnerable populations. The company became a private sector pioneer bringing together scientific innovations and industrial excellence with social and environmental responsibility for the treating and preventing malnutrition among vulnerable populations. The company deliberately designed the IP strategy with the intention to meet desired sustainability goals. In developed countries, the company uses patents to block other companies from cheaper mass production. The company licences their IP (incl. patents, trademarks, and manufacturing know-how) to franchisees in the least developed countries (LDCs)/low- and middle-income countries (LMICs), which have a high need for such products. Similarly, case company #4, another mission-driven enterprise providing energyefficient industrial, commercial and residential technological solutions has reformulated their IP strategy for a specific energy-efficient technology developed by them. The company opened (making the technology freely available for anyone to use) the previously patent-protected technology (while still maintaining ownership through patent renewals) for more significant social responsibility for other manufacturers to build energy-efficient systems. When asked about the intention behind reformulating the IP sharing approach, the response was,

² A detailed case study of case#1, a mission-driven innovative nutrition solution provider is available at https://ip4sustainability.files.wordpress.com/ 2022/12/nutriset-case-study_ip-strategy-for-humanitarian-aid.pdf.



Fig. 1. Multi-step case study data collection process overview (source: own creation).





"... If any patent is not relevant, we will just abandon it so that anyone can use it; then, there is no question of sharing it. If we are maintaining a patent and then sharing it, then definitely it is for a particular cause that it is required by the industry for bigger social responsibilities ... any old patents anyway will expire, and people will be able to use them. We don't hold on to that". [Case #4] The deliberation dimension forms a critical part of responsible IP strategy because it has a dedicated orientation towards achieving social and environmental impact rather than just economic impact. Regular business firms, i.e., those with non-sustainability-oriented innovations, may use IP deliberately for achieving more of their financial and economic goals and less commonly for social or environmental impact. They may do socially responsible activities as a part of their corporate



Fig. 3. Five responsible IP strategy dimensions (source: own creation).

social responsibility (CSR). However, to achieve a broader and faster transition towards sustainability, they need to incorporate responsible approaches for their core innovations. Case #1 in our sample could have licensed its patents to other interested multi-nationals to increase its revenue. However, the company decided to restrict access to large multinationals that could compete with and be a hindrance to the development of local players in LMIC. Hence our case examples show that intentional and deliberate IP activities targeted towards achieving sustainability are critical for sustainable development.

4.1.2. Sustainability IP alignment

The case study results indicate that companies operating with multiple SBMs (e.g., inclusive value creation, stewardship role) adopt a mix of IP strategies (e.g., licensing, free sharing) to align the IP strategy to achieve their desired sustainability goals. Clearly, this finding is not just unique to sustainability. Alignment between IP and the company's overall strategy and product architecture is essential to enhance the performance of any business [28,71–73]. Additionally, we find responsible IP strategies are designed in such a way to align not only with the conventional economic business objectives (e.g. profit and financial gains), but also with the company's sustainability mission and SBM. Cases indicate that aligning IP strategies with the sustainability mission of companies creates a positive sustainability impact.

For instance, case company #1 adopted selective IP sharing (with know-how transfer) for a stewardship model (franchise model) and patent usage agreement model (without know-how transfer) to support inclusive value creation business model. Such selective and controlled sharing promoted socio-economic development and public health in least developed and LMICs, thus primarily leveraging the company's social impact. When asked about the importance of IP strategy for sustainability, the Managing Director said,

"Clearly if we had not had an IP strategy at the beginning of 2000, we were not being able to set up local companies in many countries ... We cannot talk about development in this context without IP. It is closely linked. The sustainable devoted business model we have right now is clearly linked to the IP strategy we took." [Case #1]

One of the well-known examples of sustainability IP alignment discussed in the literature is on the concept of open education. An example being the Massive Open Online Courses (MOOC) system. This system uses open, i.e., creative common licenses intending to offer 'cost- and selection-free education to students', but simultaneously allowing for the generation of commercial revenues (through complementary purchases such as for obtaining certificates and textbooks), depending on the level of openness of the selected open license. Such a wide reach of educational knowledge wouldn't have been possible only with the conventional education system where the educational contents are not shared freely.

4.1.3. Flexibility and timing

All five cases studied exhibit flexibility in their IP strategy by adopting a combination of IP models ranging from protective, selective sharing to free sharing of IP [18,47,56]. This finding aligns with the theory of responsible innovation (RRI), and we find responsible IP strategies embed the capacity to adapt to the changing internal and external environment and implement a mix of open as well as closed IP models over time [42]. In general, strategic flexibility has been defined as the 'ability of the organisation to adapt to substantial, uncertain and fast-occurring environmental changes that have a meaningful and positive impact on the organisation's performance' [74,75] especially in fast-paced industries [76].

For instance, case #1 implemented a patent usage agreement model as a response to humanitarian stakeholders' opinions about the monopolistic nature of patents. Case #4, on the other hand, shows flexibility in IP sharing, while keeping the core technology's IP protected, the company shares the IP for developing complementary technology and building ecosystem for sustainability. Further, some of the protected IP gets shared for free. The company has also opened (making freely available for anyone to use) the previously patent-protected

Table 1

Five responsible IP strategy dimensions for the five cases studies.

Case reference	IP strategy	Responsible IP strategy dimensions							
		Intra-firm level dimensions			Ecosystem level dimensions				
		Deliberation/ intention	Sustainability alignment	Flexibility	Inclusiveness	Co-creation			
Case #1 A mission-driven innovative nutrition company offering process and product-oriented innovations	Protective and semi- open IP strategy combining different IP models: selective licensing of IP to local entrepreneurs (denies licenses to incumbents); patent usage agreement vs industrial know-how transfer as a part of the unique network model.	Company's mandate is to provide nutritional autonomy for all. The IP strategy is designed as a means to achieve the targeted sustainable development goals (SDGs 1 & 2).	Adopts two SBMs: (i) inclusive value creation through a unique network/ franchise model and (ii) a stewardship role, supported respectively by the selective IP (including industrial know-how) sharing and patent (only) usage agreement model.	Mix of different IP models in response to internal and external conditions, e.g., patent usage agreement model as a response to humanitarian stakeholder's opinions.	IP strategy is designed to offer freedom to operate and bring together the local farmers, local manufacturers, humanitarian actors, government and local beneficiaries (children and mothers) to build a local, sustainable ecosystem.	Partly through joint R&D and IP generation with research institutes (majorly in- house R&D), IP sharing for co- creation/co- production.			
Case #2 A mission-driven renewable energy incumbent offering service-oriented innovations	Protective and semi- open IP strategy including mix of IP models: Trademark dominated strategy with selective licensing and co- branding only with sustainability-focused actors; patent focused strategy less dominant.	Driven by the founder's ideology of green and sustainability with a commitment to change the way electricity is made and used (SDG 7) to replace fossil fuels with green alternatives.	Adopts the SBM - substitute with renewables and natural processes. Trademark focused IP strategy supports signalling to customers the existence of green energy alternatives.	Mix of IP models in response to internal and external conditions: e.g., patenting for own exploitation, licensing attempts if own exploitation fails, if not then let the patent lapse.	Business strategy with IP as an integral part is designed to collaborate with only those companies that meet sustainability criteria set by the company in the supply chain, inclusive of customers' feedback.	Partly through acquisitions and university collaborations (majorly in-house IP generation).			
Case #3 A mission-driven born sustainable consumer electronics company	Open IP strategy: process and product know-how mostly not patented but shared openly with interested parties via., free reports, open-source software and open hardware. Selected patenting for securing freedom to operate. Strong trademark focus to increase brand awareness for sustainable products.	IP strategy designed to target change towards sustainability not only for their own product but for the consumer electronics/ manufacturing sector widely through supply chain focus, improving work conditions, for customers improving longevity of product and enabling repairability (SDGs 9, 8, & 12).	Adopts two SBMs: a stewardship role and closing resource loops both supported by IP sharing and co- creation strategically. Environmental and Social impact is equally important as financial impact.	Mix of IP models in response to internal and external conditions: e.g., durable and high- performance product components are based on patented IP - the business chose longevity and product usability as the compromise on quality to go for pure open hardware option was too severe. Also, moved from pure open-source software for increasing usability, market acceptance and erowth	Employs a strategic sharing approach of the IP to reach out to and collaborate with stakeholders, including competitors and businesses, other NGOs and governmental organisations along the supply chain, improve the resource extraction and manufacturing for the whole sector. Open- source and open- hardware approach for enabling customers to repair products hence increase longevity	Joint development with research institutes and stakeholders (manufacturers) and shares IP with other actors.			
Case #4 A sustainability- oriented large enterprise providing affordable, environment- friendly and energy- efficient industrial, commercial and residential technological solutions	Protective, semi-open and fully open IP strategy. Selective because the company does not share the IP on the core technology, but for complementary technologies, it enters into cross-licensing, IP consortium, product partnerships, technological & industrial know-how transfers. Selected IP is also shared freely.	IP strategy of the company enables it to build an industrial ecosystem to solve environmental issues like CO ₂ emission, energy, and food and water consumption during manufacturing and commercial activities (SDGs 7, 8, 9, 12 & 17).	Adopts two SBMs, i.e. maximise efficiency, and adopt a stewardship role. Both the SBMs get supported by its' mixed IP strategy, where the company keeps the core technology protected while sharing the IP with (also in some cases, making the IP freely available) multiple stakeholders to develop an ecosystem for sustainable development.	Mix of different IP models in response to internal and external conditions: e.g., being flexible in IP sharing while keeping the core technology's IP protected, the company shares IP for complementary technology development and ecosystem building for sustainability. Also, shares some of its protected IP for free.	IP strategy is designed to enable collaboration with the customers, supply- chain partners or vendors, competitors, and also with companies in other industries. Shares the industrial and technical know-how with small and new local players to develop an ecosystem for sustainable electricity generation for the local community from food waste.	Collaborative as well as internal R&D acquires IP from outside firms and also licences permission to use other's IP; also shares own IP with others.			
Case #5 A small mission- driven tech company providing innovative solutions for urban circularity	Protective and semi- open IP strategy: Core technology protected (prefers utility models and trade secrets over patents) and used as a	IP strategy enables the company to offer high-tech urban solutions that reduce water consumption and provide efficient	The company adopts two SBMs to create a positive sustainability impact: Creating value from waste and developing scale-up	IP strategy is flexible to accommodate growing market interest: while the core technology is being protected, the	IP strategy enables to build a reputation of "attractive collaborator", which helps the company to access new markets	Mostly in-house R&D, recently increasing in focus on collaborative, joint R&D with universities, research institutes, industry (continued on next page)			

7

Table 1 (continued)

Case reference	IP strategy	Responsible IP strategy dimensions					
		Intra-firm level dimensions			Ecosystem level dimensions		
		Deliberation/ intention	Sustainability alignment	Flexibility	Inclusiveness	Co-creation	
	platform to collaborate with other tech companies. Currently considering expanding their knowledge-sharing through licensing.	ways for water recycling and reuse, promoting ways for retrieving monetary value from municipal waste. (SDGs 7,9,11,12,& 17).	solutions for sustainability supported by the protective IP and licensing approaches.	company is developing new apps and products with partners in Europe and Asia interested in integration and adoption of the technology.	and bring together local industrial partners, municipalities, start- ups, real estate companies, innovators and researchers to work together towards sustainability targets.	partners. IP sharing through cross- licensing	

technology (while still maintaining ownership through patent renewals) for more significant social responsibility for other manufacturers to build energy-efficient systems.

"We had developed a special energy-efficient technology ... for home applications. Previously, it was our proprietary technology, but over the period, it was observed that it is a very efficient system that can help many households, and other manufacturers can also develop communications with these kinds of protocols. Subsequently, we opened this technology for the benefit of society ... The criteria to decide whether to open a patent or not is not the relevancy of the patent" [Case #4]

The timing of their IP asset types forms another aspect of flexibility that became evident from case companies. For example, companies find certain IP assets (e.g. trademarks and know-how) more valuable in the early and matured stage of their business and other IP assets (e.g. patents) more valuable during the growth phase. Such timing of IP assets gets evidenced in regular business as well (not only sustainabilityoriented) and sometimes form a natural part of an IP strategy evolution [42].

4.2. Ecosystem level dimensions

Ecosystem level category includes two dimensions namely inclusiveness, and co-creation.

4.2.1. Inclusiveness

All the five case studies exhibit inclusiveness as part of how they use IP for sustainability. The finding again concurs with prior literature. Stakeholder inclusiveness has been shown to promote sustainability in different sectors [77,78]. As a part of their responsible IP strategy, the case companies proactively embed stakeholder ecosystem building as a prime concern while developing sustainability practices both within their organisation and among stakeholders within the ecosystem (e.g., value chain or supply chain partners).

For example, case company #2 engages in co-branding, collaboration and partnerships, thus being inclusive and boundary spanning, but strategically only with actors satisfying specific sustainability criteria and denying IP usage rights to unsustainable actors. The company uses the company trademark as an assessment tool for selecting partners, i.e. as a contractual entry barrier for their ecosystem admitting only those companies that meet sustainability criteria set by the case company for supply chain. If any partner company does not meet the sustainability criteria at any time, the company reserves the right to terminate the trademark licensing agreement.

"... standard [trademark] licence agreement will give us the right to terminate if they [partners] are engaged in activity that is derogatory, unlawful ... " [Case #2]

With the selective IP approach model, the company has built a renewable energy ecosystem with hundreds of partners. In addition to supply chain partners, the company has also used its trademarks as a valuable signalling tool to create awareness about sustainable alternatives and incorporated feedback from customers in the business operations.

"... talking about trademarks, company names, and domain names as protective initially in the sense of gathering an exclusivity ... the use of a mark like that does in fact raise awareness of green products, raise awareness of green services, alerts people to the fact that they don't have to buy non-green brand ... " [Case #2]

There are many examples in different industries of how businesses use an inclusiveness approach for functionality of technology. They share IP assets with stakeholders such as supply chain members and competitors and integrate user perspectives and different types of users, e.g. on data transmission technologies in the electronic sector, and to achieve safety standards such as airbags in automotive that are suitable for people of different sizes and weight. The cases reveal that this approach appears also suitable to build ecosystems for environmental and social sustainability solutions.

4.2.2. Co-creation

Responsible IP strategies are boundary spanning, accessing relevant IP from external sources and sharing own IP with others in need to promote sustainability benefits not only within the organisation but also to the external market, stakeholders and society at large. All five cases exhibit co-creation through boundary-spanning behaviours in one form or another. Such boundary-spanning activities promote the co-production of knowledge and co-designing needed for sustainability transformative changes [69,70].

Case #3, for instance, mentions open sharing of IP as a precondition for building strong partnerships with stakeholders. The value of sharing, however, is outbound as well as inbound. The IP developments were all collaboratively created with others:

"... so basically almost everything and everything in here, everything that we build in the impact innovation is developed with others. I don't think we have something that we really came up with only ourselves. Which would probably also be impossible. because you always have to talk to other experts" [Case #3]

When asked if the company regrets sharing their IP openly instead of restricted use, the interviewee responded:

"Well, I don't think we have a choice here because we basically have to build a partnership. And when you build a partnership, when there's not, well, you know, you have to actually convince people to want to join this partnership." [Case #3]

Case company #4, as their IP strategy, protects the core energy-

efficient technology while entering into diverse collaborations with supply-chain partners, competitors, and companies in unrelated industries to develop complementary technologies and sub-systems around the core technology.

"... we and our close competitors for that field have an industry consortium ... where 5–6 companies come together and we crosslicense our IP so that each one of us can use that IP and we can develop a next generation technology by holding hands with each other ..." [Case #3]

Similarly, case company #5, as part of the IP strategy, uses IP to protect the core technology and use IP as a platform for joint R&D of new apps and products with other industries, and cross-licensing to stakeholders interested in integrating and adopting the technology. IP gets mainly developed through in-house generation, but recently the company shows increased focus on open innovation. Several newly developed technologies have been developed jointly with partners inside and outside of the industry through product partnerships and cross-licensing agreements.

"At the beginning, I wanted to be sure that our part, our core technology, is good and safe, and protected and then we can go out to find partners and ... good cooperation where our technology is acknowledged as an important part of something we are going to develop together". [Case #5]

Research on innovation, especially on open innovation, has provided insights into the benefits of co-creation to develop and improve more robust solutions that serve societal needs and at the same time increase the speed of diffusion [22,23,79].

4.3. Validation of the responsible IP strategy dimensions: a case of a company transitioning to sustainability

To validate the dimensions of responsible IP strategy resulting from the case studies, we applied the dimensions to a sixth case company transitioning to sustainability. In other words, the company has shifted from being a traditional (bottom line focused) business to a sustainability-oriented one. The company is a large B2B multinational (over 130 years old) started in the energy sector but later expanded into other related industries such as power, electronics, and cold chain. The company incorporated sustainability as a part of its core business objective in 2019. We specifically selected a company in transition for validation as in-transition companies form a good example to study the dimensions before and after change towards sustainability orientation.

The company underwent two major changes. First, the company restructured the business units in 2015, with sustainability not as an explicitly stated business objective, but as implicit transition evidenced through their shifting focus towards sustainable solutions (e.g. energy efficiency, CO_2 emission focus). The company made sustainability a core part of the business objective in 2019. Prior to 2015, the company shifted from cost reduction to energy efficiency at the overall system level in 2013. Hence, we consider the time window from 2013 onwards as the company's sustainability transition period and compare the dimensions before and during the transition period as depicted in Fig. 4.

Along the sustainability IP alignment dimension, the company has clearly reoriented the business model from a predominantly product development and integration-based business model to an ecosystem actor-focused business model around early 2010s and recently to outcome-based and service-focused business model since 2019. While the company has always aligned the IP strategy to its business strategy (e.g., IP protection to support own product development and integration), strategic thinking about IP to support sustainability is evidenced only during the transition period from 2015, and more strongly since 2019. Accordingly, we observe a shift in the IP focus from a legal asset to protect inventions for own usage. We also witness the company using IP as a business tool for enabling collaboration with other industry players, start-up hubs, new technology entities, and even with customers to support the company's sustainability business objectives.

Along the deliberation dimension, the company's sustainability impact contribution in terms of energy efficiency and reduced CO_2 emission started once the company intentionally made sustainability its core business objective and used IP to engage with system-level actors to achieve energy efficiency. Even before the transition period, the company engaged in several CSR activities such as providing monetary and



Fig. 4. Responsible IP strategy dimensions of a company transitioning to sustainability (source: own creation).

in-kind support worldwide in education, public health, training, skill development and several other philanthropic activities. However, the role of IP for sustainability was limited. The negative impact on the environment from core business started reducing once the company made deliberate actions to steer the core business and IP activities towards sustainability impacts.

For the flexibility and timing dimensions, the case shows a mix of IP models, namely, protecting IP for own use and IP sharing for mostly value creation for the company through collaboration. However, this flexibility started a few years before the transition period in response to the need to accelerate its innovation process (around 2012). The company, though does not actively engage in out-licensing, has reported a willingness to monetize the IP in future through sharing (licensing). This probably indicates the need for further consideration of the dimension by the company for accelerating the transition.

The inclusiveness dimension gets clearly evident from the above descriptions as well as the company's engagement more widely with start-up hubs, new technology entities and end customers to involve them in the company's sustainability journey. Specifically, for the cold storage business in the food industry, the company engages with players ranging from farms to retail, in order to ensure food quality and safety.

Along the co-creation dimension, the company has moved away from only relying on their own IP creation. The deliberate focus on sustainability since 2019 also pushed the company to co-create not only with industry partners but also with end users and customers, indicating also the interconnectedness among the dimensions.

The case also demonstrated that sustainability journeys are not being free of obstacles. To quote one example, the company started engaging in collaborative innovations in the beginning without careful attention to IP terms in the collaboration agreement, particularly the ownership of the foreground IP (IP generated from the collaboration). The collaborator, an original equipment manufacturer (OEM), went and filed for patent protection independently for the foreground IP. Consequently, the case company lost the opportunity to license the IP to other OEMs in the market. With the lesson learned from the first collaboration, in subsequent collaborations, the company has been carefully drafting the IP terms in the legal agreement to protect any possibilities to offer solutions to multiple market players. Understanding the conflicts and challenges that arise in implementing the dimensions deserves separate investigation and can be worthy of future research. Here, we present some insights about a possible challenge in implementing the dimensions and moving towards building a responsible IP strategy.

The above case example demonstrate that the dimensions identified for the responsible IP strategy have certain characteristics that nonsustainability performing entities may lack, as evidenced from the pretransition status for the company.

4.4. Defining responsible IP strategy (R–IPS)

Following the results and identification of the five dimensions based on the case study data, we propose to define the notion of 'responsible IP strategy' (R–IPS) as follows:

"A responsible IP strategy is an IP strategy that deliberately aligns with the organization's sustainability mission; is inclusive (in terms of IP sharing) of sustainability promoting stakeholders; and spans the boundary to enable knowledge co-creation, such as for common purpose technologies. Such IP strategy considers the different impact dimensions (e.g. social, environmental, and economic implications) and societal expectations (e.g. in its value definition) to flexibly evolve over time to best support long-term sustainability transitions."

We also find inter-dependence among the five dimensions of the responsible IP strategy. For example, a firm with a sustainability mission also shows the tendency to strategize their IP sharing and usage intentionally to promote sustainable development. To give a specific example, case #1 embedded inclusiveness (i.e., IP strategy targeted towards local players) and flexibility (different IP models to cater to different IP users to increase sustainability impact) in the IP strategy, primarily because of its sustainability orientation (sustainability alignment) of providing nutrition for all. Evidently, intra-firm sustainability orientation forms a pre-condition for firms to use IP responsibly to create sustainability within the wider innovation ecosystem.

5. Conclusions and contributions

This research contributes to the broader debate on the role of IP and its strategic management for sustainability. The paper provides evidence-based findings about IP strategies that sustainability-focused companies adopt to balance their societal responsibility, mission and business goals without compromising (too much) on their environmental and/or social impact at their internal operational level, external market level, and wider sectoral level. Based on these findings, we contribute by conceptualising and defining 'responsible IP strategies'.

Our research identifies five dimensions that are integral to responsible IP strategies, namely deliberation/intention, sustainability IP alignment, flexibility and timing, inclusiveness, and co-creation to take into account for a company desiring to formulate IP strategy for achieving their company's mission without compromising on environmental and/or societal impact.

We also find that while RRI provides a reasonable basis for investigating responsible IP strategies, the dimensions of RRI, namely reflexivity, anticipation, inclusiveness, and responsiveness have limitations in fully explaining what constitutes a responsible IP strategy. At the firm level, when formulating a responsible IP strategy for sustainability, companies incorporate the reflexivity and responsiveness dimensions of RRI through the two dimensions, namely deliberation/intention and flexibility and timing, respectively. For the anticipation dimension, the RRI suggests creating appropriate solutions and policies. Adapting the same for the firm-level IP strategy, we find that companies create appropriate IP and sharing practices by aligning them with their sustainable business objectives, as captured through the sustainability IP alignment dimension. For the inclusive dimensions, however, responsible IP strategy goes beyond what RRI suggests as inclusive. We find companies adopting inclusiveness in IP strategy not only listen to the public and the stakeholders as the RRI suggests but use IP assets as a tool as well to steer the stakeholders and public towards sustainable practices and behaviour. Co-creation emerges as a dimension of the responsible IP strategy but does not get adequately captured as a part of RRI.

The concept of a responsible IP strategy provides a framework for managerial decision making in IP strategy related matters linked to sustainability. The proposed dimensions, if turned into a scoring tool, i. e. scored on a scale from high to low, can also be used as an assessment (diagnostics) tool for companies to assess the level of responsibility of their IP strategy and to indicate areas of improvement in formulating an even better responsible IP strategy to maximise the company's contribution towards sustainable development goals.³ In terms of policy implications, the five dimensions indicate that responsible IP strategies require support beyond what the current IPR systems offer and emphasise the need for strategic focus on policy initiatives targeted at the use and particuarlly sharing of IP for sustainability.

The paper is not free of limitations and opens multiple avenues for further research. Though we have carefully selected the case companies, our current investigation is based on five case studies from varied industries and future research could focus on validating this framework using more case studies or quantitative data from a homogeneous industry setting. Furthermore, the external validity is limited in the sense

³ An example of a toolkit developed by the IPACST project team for IP asset assessment for increasing sustainability impact is available at https://ip4sustain ability.org/knowledge-briefs/.

we validate the framework using only one case company in its transition period. Future research applying the framework to multiple case studies can further improve the trustworthiness and rigor of the framework. We do not include contextual differences in responsible IP strategies, and that forms another part of the limitation. Future investigations for contextual differences such as variation across industries, technology types, firm sizes, and sustainability orientations (social vs environmental) will add to the insights in the paper. Further, for enhanced managerial implications, the contributions from the paper can be used as a basis for developing a managerial tool for formulating responsible IP strategies.

CRediT authorship contribution statement

Pratheeba Vimalnath: Conceptualization, Investigation, Formal analysis, Writing – original draft, Writing – review & editing. Frank Tietze: Conceptualization, Writing – original draft, Writing – review & editing. Elisabeth Eppinger: Investigation, Writing – original draft, Writing – review & editing. Akriti Jain: Investigation, Writing – original draft, Writing – review & editing. Anjula Gurtoo: Writing – original draft, Writing – review & editing. Maximilian Elsen: Writing – review & editing.

Declaration of interest

The authors declare the following financial interests/personal relationships which may be considered as potential competing interests

Data availability

The data in anonymized form is available upon request.

References

- B.S. Silvestre, D.M. Ţîrcă, Innovations for sustainable development: moving toward a sustainable future, J. Clean. Prod. 208 (2019) 325–332.
- [2] R. Walz, M. Pfaff, F. Marscheider-Weidemann, S. Glöser-Chahoud, Innovations for reaching the green sustainable development goals-where will they come from? Int. Econ. Econ. Pol. 14 (3) (2017) 449–480.
- [3] M. Urbaniec, Towards sustainable development through eco-innovations: drivers and barriers in Poland, Econ. Sociol. 8 (4) (2015) 179.
- [4] F. Tietze, P. Vimalnath, L. Aristodemou, J. Molloy, Crisis-critical intellectual property: findings from the COVID-19 pandemic, IEEE Trans. Eng. Manag. 69 (5) (2020) 2039–2056.
- [5] T. Schiederig, F. Tietze, C. Herstatt, Green innovation in technology and innovation management–an exploratory literature review, R D Manag. 42 (2) (2012) 180–192.
- [6] W. Ali, J. Wen, H. Hussain, N.A. Khan, M.W. Younas, I. Jamil, Does green intellectual capital matter for green innovation adoption? Evidence from the manufacturing SMEs of Pakistan, J. Intellect. Cap. 22 (5) (2021) 868–888.
- [7] G. Cainelli, V. De Marchi, R. Grandinetti, Does the development of environmental innovation require different resources? Evidence from Spanish manufacturing firms, J. Clean. Prod. 94 (2015) 211–220.
- [8] B.H. Hall, C. Helmers, Innovation and diffusion of clean/green technology: can patent commons help? J. Environ. Econ. Manag. 66 (1) (2013) 33–51.
- [9] V. Verendel, Tracking artificial intelligence in climate inventions with patent data, Nat. Clim. Chang. 13 (2023) 40–47, https://doi.org/10.1038/s41558-022-01536restriction of the state of th
- [10] DeepMind AI Reduces Google Data Centre Cooling Bill by 40%, 2016. https:// www.deepmind.com/blog/deepmind-ai-reduces-google-data-centre-cooling-bill-b y-40. (Accessed 2 December 2022).
- [11] E.G. Hansen, F. Grosse-Dunker, Sustainability-oriented Innovation, Encyclopedia of Corporate Social Responsibility, Heidelberg, Germany, 2012.
- [12] J. Klewitz, E.G. Hansen, Sustainability-oriented innovation of SMEs: a systematic review, J. Clean. Prod. 65 (2014) 57–75.
- [13] E. Eppinger, A. Jain, P. Vimalnath, A. Gurtoo, F. Tietze, R.H. Chea, Sustainability transitions in manufacturing: the role of intellectual property, Curr. Opin. Environ. Sustain. 49 (2021) 118–126.
- [14] F. Tietze, J. Sternkopf, E. Eppinger, P. Vimalnath, IP Strategies for Sustainability when Does Open IP Increase Sustainable Impact? 1st Annual International Conference of the IEEE Technology and Engineering Management Society (TEMSCON 2017), California, USA, 2017, pp. 173–178. June 8-10, 2017.
- [15] A. Moerchel, F. Tietze, L. Aristodemou, P. Vimalnath, Identifying crisis-critical intellectual property challenges during the covid-19 pandemic, Vol. 2020, No. 1, in: Academy of Management Proceedings, Academy of Management, NY 10510, 2021, p. 15224. Briarcliff Manor.

- [16] J.L. Contreras, M. Eisen, A. Ganz, M. Lemley, J. Molloy, D.M. Peters, F. Tietze, Pledging intellectual property for COVID-19, Nat. Biotechnol. 38 (2020) 1146–1149.
- [17] B. Hall, C. Helmers, M. Rogers, V. Sena, The choice between formal and informal intellectual property: a review, J. Econ. Lit. 52 (2) (2014) 375–423.
- [18] P. Vimalnath, F. Tietze, E. Eppinger, J. Sternkopf, Closed, semi-open, or fully-open? Towards an intellectual property strategy typology, Vol. 2020, No. 1, in: Academy of Management Proceedings, Academy of Management, NY 10510, 2020, p. 22070. Briarcliff Manor.
- [19] T. Wang, F. Tietze, R. Phaal, Intellectual property strategy development through roadmapping, in: Proceedings of the R&D Management Conference, 2018. Milan, Italy(vol. 30).
- [20] O. Granstrand, M. Holgersson, Innovation ecosystems: a conceptual review and a new definition, Technovation 90 (2020), 102098.
- [21] J. Stilgoe, R. Owen, P. Macnaghten, Developing a framework for responsible innovation, in: The Ethics of Nanotechnology, Geoengineering and Clean Energy, Routledge, 2020, pp. 347–359.
- [22] R.V. Schomberg, Prospects for technology assessment in a framework of responsible research and innovation, in: Technikfolgen Abschätzen Lehren, VS Verlag für Sozialwissenschaften, 2012, pp. 39–61.
- [23] R. Owen, J. Stilgoe, P. Macnaghten, M. Gorman, E. Fisher, D. Guston, A framework for responsible innovation, in: Responsible innovation: managing the responsible emergence of science and innovation in society 31, 2013, pp. 27–50.
- [24] H. Delerue, A. Lejeune, Managerial secrecy and intellectual asset protection in SMEs: the role of institutional environment, J. Int. Manag. 17 (2) (2011) 130–142.
- [25] M. Holgersson, M.W. Wallin, The patent management trichotomy: patenting, publishing, and secrecy, Manag. Decis. 55 (2017) 1087–1099.
- [26] R.H. Pitkethly, Intellectual property strategy in Japanese and UK companies: patent licensing decisions and learning opportunities, Res. Pol. 30 (3) (2001) 425–442.
- [27] C. Oh, S. Matsuoka, Complementary approaches to discursive contestation on the effects of the IPR regime on technology transfer in the face of climate change, J. Clean. Prod. 128 (2016) 168–177.
- [28] M. Reitzig, Strategic management of intellectual property, MIT Sloan Manag. Rev. 45 (3) (2004) 35–40.
- [29] Y. Kim, N.S. Vonortas, Determinants of technology licensing: the case of licensors, Manag. Decis. Econ. 27 (4) (2006) 235–249.
- [30] G.P. Pisano, D.J. Teece, How to capture value from innovation: shaping intellectual property and industry architecture, Calif. Manag. Rev. 50 (1) (2007) 278–296.
- [31] J.M. Lee, R. Narula, J. Hillemann, Unraveling asset recombination through the lens of firm-specific advantages: a dynamic capabilities perspective, J. World Bus. 56 (2) (2021), 101193.
- [32] R. Lema, A. Lema, Technology transfer? The rise of China and India in green technology sectors, Innovat. Dev. 2 (1) (2012) 23–44.
- [33] Y. Benkler, Power and Productivity: Institutions, Ideology, and Technology in Political Economy. Danielle Allen, Yochai Benkler, Leah Downey, Rebecca Henderson, & Josh Simon, A Political Economy Of Justice, University of Chicago Press 2022, 2022.
- [34] G. Dosi, L. Marengo, C. Pasquali, How much should society fuel the greed of innovators?: on the relations between appropriability, opportunities and rates of innovation, Res. Pol. 35 (8) (2006) 1110–1121.
- [35] F.W. Geels, J. Schot, Typology of sociotechnical transition pathways, Res. Pol. 36 (3) (2007) 399–417.
- [36] P. Söderholm, The green economy transition: the challenges of technological change for sustainability, Sustain. Earth 3 (1) (2020) 1–11.
- [37] N. Corrocher, F. Malerba, A. Morrison, Technological regimes, patent growth, and catching-up in green technologies, Ind. Corp. Change 30 (4) (2021) 1084–1107.
- [38] C. Henry, J.E. Stiglitz, Intellectual property, dissemination of innovation and sustainable development, Global Pol. 1 (3) (2010) 237–251.
- [39] N. Barbieri, A. Marzucchi, U. Rizzo, Knowledge sources and impacts on subsequent inventions: do green technologies differ from non-green ones? Res. Pol. 49 (2) (2020), 103901.
- [40] A. Potter, S. Graham, Supplier involvement in eco-innovation: the co-development of electric, hybrid and fuel cell technologies within the Japanese automotive industry, J. Clean. Prod. 210 (2019) 1216–1228.
- [41] G. Roos, J. Roos, Measuring your company's intellectual performance, Long. Range Plan. 30 (3) (1997) 413–426.
- [42] P. Vimalnath, F. Tietze, A. Jain, A. Gurtoo, E. Eppinger, M. Elsen, Intellectual property strategies for green innovations-An analysis of the European Inventor Awards, J. Clean. Prod. 377 (2022), 134325, https://doi.org/10.1016/j. jclepro.2022.134325.
- [43] C.M. Jardon, A. Dasilva, Intellectual capital and environmental concern in subsistence small businesses, Manag. Environ. Qual. Int. J. (2017).
- [44] J.J. Ferreira, C.I. Fernandes, F.A. Ferreira, Technology transfer, climate change mitigation, and environmental patent impact on sustainability and economic growth: a comparison of European countries, Technol. Forecast. Soc. Change 150 (2020), 119770.
- [45] D. Alonso-Martínez, Social progress and international patent collaboration, Technol. Forecast. Soc. Change 134 (2018) 169–177.
- [46] D. Matricano, Economic and social development generated by innovative startups: does heterogeneity persist across Italian macro-regions? Econ. Innovat. N. Technol. (2020) 1–18.
- [47] A. Jain, A. Gurtoo, Can the restrictive boundaries of intellectual property create a larger social impact? Curr. Sci. 120 (11) (2021) 1667.
- [48] C.M. Correa, Ownership of knowledge: the role of patents in pharmaceutical R&D, Bull. World Health Organ. 82 (2004) 784–787.

P. Vimalnath et al.

- [49] D. Libaers, D. Hicks, A.L. Porter, A taxonomy of small firm technology commercialization, Ind. Corp. Change 25 (3) (2016) 371–405.
- [50] J.L. Contreras, B.H. Hall, C. Helmers, Pledging patents for the public good: rise and fall of the eco-patent commons, Houst. Law Rev. 57 (2019) 61.
- [51] A. Gambardella, A.M. McGahan, Business-model innovation: general purpose technologies and their implications for industry structure, Long. Range Plan. 43 (2–3) (2010) 262–271.
- [52] W. Lesser, Intellectual property rights and concentration in agricultural biotechnology, AgBioforum 1 (2) (1998) 56–61.
- [53] H. König, P. Dorado-Morales, M. Porcar, Responsibility and intellectual property in synthetic biology: a proposal for using Responsible Research and Innovation as a basic framework for intellectual property decisions in synthetic biology, EMBO Rep. 16 (9) (2015) 1055–1059.
- [54] V. Munoz, F. Visentin, D. Foray, P. Gaulé, Can medical products be developed on a non-profit basis? Exploring product development partnerships for neglected diseases, Sci. Publ. Pol. 42 (3) (2015) 315–338.
- [55] G. George, A.M. McGahan, J. Prabhu, Innovation for inclusive growth: towards a theoretical framework and a research agenda, J. Manag. Stud. 49 (4) (2012) 661–683.
- [56] E. Eppinger, P. Tinnemann, Technology transfer of publicly funded research results from academia to industry: societal responsibilities? in: J. van den Hoven, H. Romijn, T. Swierstra, N. Doorn (Eds.), Responsible Innovation 1, Innovative Solutions for Global Issues Springer, Dordrech, Netherlands, 2014, pp. 67–88 (Springer, Dordrecht).
- [57] P. Beyer, Developing socially responsible intellectual property licensing policies: non-exclusive licensing initiatives in the pharmaceutical sector, in: Research Handbook on Intellectual Property Licensing, Edward Elgar Publishing, 2013, pp. 227–256.
- [58] A. Brown, IP, regulation and the enabling state, strathclyde conference 2005. https: //era.ed.ac.uk/bitstream/handle/1842/2523/27_brownipregulationsep05.pdf?se quence=1, 2005.
- [59] K.M. Eisenhardt, Building theories from case study research, Acad. Manag. Rev. 14 (4) (1989) 532–550.
- [60] R.K. Yin, Case Study Research: Design and Methods, 5, sage, 2009.
- [61] S.V. de Freitas Netto, M.F.F. Sobral, A.R.B. Ribeiro, G.R.D.L. Soares, Concepts and forms of greenwashing: a systematic review, Environ. Sci. Eur. 32 (1) (2020) 1–12.
- [62] M.A. Delmas, V.C. Burbano, The drivers of greenwashing, Calif. Manag. Rev. 54 (1) (2011) 64–87.
- [63] N.M. Bocken, S.W. Short, P. Rana, S. Evans, A literature and practice review to develop sustainable business model archetypes, J. Clean. Prod. 65 (2014) 42–56.
 [64] H.F. Hsieh, S.E. Shannon, Three approaches to qualitative content analysis, Qual.
- Health Res. 15 (9) (2005) 1277–1288. [65] L. Rourke, T. Anderson, Validity in quantitative content analysis, Educ. Technol.
- Res. Dev. 52 (1) (2004) 5–18.
 [66] A.K. Shenton, Strategies for ensuring trustworthiness in qualitative research projects, Educ. Inf. 22 (2) (2004) 63–75.
- [67] E.G. Guba, Criteria for assessing the trustworthiness of naturalistic inquiries, Ectj 29 (2) (1981) 75–91.
- [68] D.M. Iwaniec, E.M. Cook, O. Barbosa, N.B. Grimm, The framing of urban sustainability transformations. Sustainability 11 (3) (2019) 573.
- [69] G.G. Page, R.M. Wise, L. Lindenfeld, P. Moug, A. Hodgson, C. Wyborn, I. Fazey, Codesigning transformation research: lessons learned from research on deliberate practices for transformation, Curr. Opin. Environ. Sustain. 20 (2016) 86–92.
- [70] A.M. Meadow, D.B. Ferguson, Z. Guido, A. Horangic, G. Owen, T. Wall, Moving toward the deliberate coproduction of climate science knowledge, Weather Clim. Soc. 7 (2) (2015) 179–191.
- [71] J. Henkel, C.Y. Baldwin, W. Shih, IP modularity: profiting from innovation by aligning product architecture with intellectual property, Calif. Manag. Rev. 55 (4) (2013) 65–82.
- [72] M.J. Lynskey, Aligning strategy and intellectual property to maximise business value: a proposal for new technology-based firms, Int. J. Intellect. Property Manag. 3 (4) (2009) 301–325.
- [73] R. Hernández-Chea, P. Vimalnath, N. Bocken, F. Tietze, E. Eppinger, Integrating intellectual property and sustainable business models: the SBM-IP canvas, Sustainability 12 (21) (2020) 8871.
- [74] D.A. Aaker, B. Mascarenhas, The need for strategic flexibility, J. Bus. Strat. 5 (2) (1984) 74.
- [75] N. Worren, K. Moore, P. Cardona, Modularity, strategic flexibility, and firm performance: a study of the home appliance industry, Strat. Manag. J. 23 (12) (2002) 1123–1140.
- [76] S. Nadkarni, V.K. Narayanan, Strategic schemas, strategic flexibility, and firm performance: the moderating role of industry clockspeed, Strat. Manag. J. 28 (3) (2007) 243–270.
- [77] I.M. Herremans, J.A. Nazari, F. Mahmoudian, Stakeholder relationships, engagement, and sustainability reporting, J. Bus. Ethics 138 (3) (2016) 417–435.
- [78] D. Lingenfelder, Stakeholder inclusiveness in sustainability reporting by mining companies listed on the Johannesburg securities exchange, Afr. J. Bus. Ethics 5 (1) (2011).
- [79] E. Eppinger, How open innovation practices deliver societal benefits, Sustainability 13 (3) (2021) 1431, https://doi.org/10.3390/su13031431.

Dr Pratheeba Vimalnath is a Lecturer in Innovation, Intellectual Property and Sustainability at the University of Exeter Business School, U.K. She teaches and researches in the area of strategic innovation and intellectual property management for sustainability and circular economy. Prior to joining the University of Exeter, she worked as a senior research associate at the University of Cambridge, U.K., and before that as an Oxford Martin Fellow with the University of Oxford, U.K. She has published her work in various international journals, delivered several invited talks and has experience working for projects funded under the Economic and Social Research Council (ESRC), British Council's Going Global Partnership Program, European Commission, Horizon 2020, and NIAS(India)-United Nations Development Program (UNDP) policy research initiative. She has three years of industrial experience in telecommunication and banking domains. She holds a bachelor's degree in computer science and engineering from the Government College of Technology (Anna University), Coimbatore, India, and a master's degree in technology management and a doctorate degree in intellectual property management from the Indian Institute of Science, India.

Prof Frank Tietze is Professor of Innovation Engineering at the University of Cambridge, Cambridge, U.K. At the Institute for Manufacturing (IfM) and within the Centre for Technology Management, he leads the Innovation and Intellectual Property Management (IIPM) Laboratory. He is a steering group member of Cambridge Global Challenges, affiliated to the Cambridge Centre for Intellectual Property and Information Law and a Member of the Innovation and the Intellectual Property Research Group, Chalmers University of Technology, Sweden. His research has been published widely in leading international journals. He is a Departmental Editor for the IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT, an Editorial Board Member of the World Patent Information, and an Editorial Review Board Member of LES Nouvelles.

Prof Elisabeth Eppinger is Professor for sustainability and textile technology at the University of Applied Sciences for Technology and Economics Berlin (HTW Berlin). She led the interdisciplinary research project Intellectual Property Models for Accelerating Sustainability Transitions (IPACST) with researchers from Freie University Berlin, Indian Institute of Science, University of Cambridge and University of Lund (www.ip4sustainability.org). Over the last decades, she has been researching sustainable innovation, technology transitions and innovation collaboration with a focus on the role of intellectual property rights. She holds a PhD in innovation studies, a master in Science and Technology Studies, and an engineering degree in textile technology. Prior to conducting academic research, she gained experience in R&D in the chemical and textile industry sectors in Japan, Australia, and the Netherlands.

Dr Akriti Jain works as an Assistant Professor in Strategy & General Management at the International Management Institute (IMI), New Delhi, India. Before joining IMI, she worked as a post-doctoral fellow at the Centre for Society and Policy, Indian Institute of Science, Bengaluru, India, for the project titled, "Intellectual Property Models to Accelerate Sustainability Transitions (IPACST). She also worked as a Fellow and Consultant at the CUTS Institute for Regulation and Competition (CIRC), New Delhi, to research promoting Innovation and competition for efficient use of Intellectual Property (IP) in India. Dr Jain holds a doctorate in Economics of Intellectual Property (IP) in India. Dr Jain holds a doctorate in Economics of Intellectual Property (IP) is from the Indian Institute of Technology (IIT) – New Delhi, India. Previously, she held several teaching and corporate communication positions in India's Science and Technology institutes and taught graduate and master-level courses (subjects) like Managerial Economics, Industrial Engineering, International Management, and Science, Technology and Innovation Policy (STIP). She holds a Master's and a Bachelor's degree in Business Administration from India.

Prof Anjula Gurtoo is a Professor at the Department of Management Studies and Chairs the Centre for Society and Policy at the Indian Institute of Science, Bangalore. She is leading several national and international research projects in the domain of public and urban systems. She obtained her master's in psychology from the University of Delhi in 1992 and pursued doctoral research at the Indian Institute of Management, Ahmedabad in 1998. She has been a Fellow of the International University of Japan, Niigata (2015), University of Leeds, United Kingdom (2006), and Social Science Research Council, Washington DC, USA (2003). She has authored three books and over 70 research articles in leading international refereed journals. She is a recipient of various national and international recognitions. She also has expert membership of various government and academic committees.

Maximilian Elsen is a PhD candidate at the Institute for Manufacturing, university of Cambridge, under the supervision of Dr Frank Tietze. His research focuses on using datadriven approaches in the field of technology and innovation management. He is interested in the valuation of technologies and innovation to enhance decision making processes. Prior to starting his PhD, Max earned a B.Sc. and M.Sc. in Mechanical Engineering and Business Administration from RWTH Aachen University as well as a M.Sc. in Mechanical Engineering from Tsinghua University. He further holds a M.Sc. in Business Analytics from University College London and was a visiting student at the Massachusetts Institute of Technology.